Author’s response to reviews

Title: Youth Working in Tobacco Farming: Effects on Smoking Behavior and Association with Health Status

Authors:

Ethel Alderete (alderetew@gmail.com)
Jennifer Livaudais-Toman (Jennifer.Toman@ucsf.edu)
Celia Kaplan (Celia.Kaplan@ucsf.edu)
Steven Gregorich (Steven.Gregorich@ucsf.edu)
Raúl Mejía (raulmejia57@gmail.com)
Eliseo Perez-stable (eliseo.perez-stable@nih.gov)

Version: 2 Date: 27 Dec 2019

Author’s response to reviews:

Dear Editors,

Thank you for the letter dated December 9 that our manuscript # 19-03729R1 entitled “Youth Working in Tobacco Farming: Effects on Smoking Behavior and Association with Health Status” is potentially acceptable for publication as a Research article in the BMC Public Health pending revisions. We have added the requested disclaimer that the funding bodies had no role in study design, data collection and analysis, and interpretation and write up of results. The manuscript and tables are in one file with no color highlights or track changes. The reviewer comments were all previously addressed. We hope that it will now be acceptable for publication in BMC Public Health.

We have now updated and inserted the Declarations section.

All authors have read and agreed to the content, are accountable for the content, approved the manuscript, and declared that they have no competing interests. The manuscript conforms to the journal’s editorial policies and has not been previously published in any format nor submitted to another journal. The authors agree to the terms of the BioMed Central Copyright and License Agreement. We are aware of the fees if accepted for publication.

We appreciate your consideration.

Sincerely yours,

Eliseo J. Perez-Stable, MD
For the authors